Math 205c Spring 2005 R. Schultz

UPDATED GENERAL INFORMATION — JANUARY 9, 2004

The exercises from the homework file recommended for completion by April 6 were as follows:

I.1: 1–7 I.2: 1–3 I.3: 1–2 I.4: 2–4

The exercises from the homework file recommended for completion by April 13 are as follows:

II.1: 1–8 (review of multivariable calculus) II.1: 11 II.2: 1–3, 5, 9, 11, 12 II.3: 1, 2 II.4: 1–5 A tentative date for the midterm examination is Wednesday, May 4. This will also probably be the due date for the first written assignment.